

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Bace, Edward; Ferreira, Ana

Article Regulation's influence on EU banking efficiency: An evaluation post crisis

Cogent Economics & Finance

Provided in Cooperation with: Taylor & Francis Group

Suggested Citation: Bace, Edward; Ferreira, Ana (2020) : Regulation's influence on EU banking efficiency: An evaluation post crisis, Cogent Economics & Finance, ISSN 2332-2039, Taylor & Francis, Abingdon, Vol. 8, Iss. 1, pp. 1-21, https://doi.org/10.1080/23322039.2020.1838735

This Version is available at: https://hdl.handle.net/10419/270008

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



WWW.ECONSTOR.EU

https://creativecommons.org/licenses/by/4.0/

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.







Cogent Economics & Finance

ISSN: (Print) (Online) Journal homepage: https://www.tandfonline.com/loi/oaef20

Regulation's influence on EU banking efficiency: An evaluation post crisis

Edward Bace & Ana Ferreira |

To cite this article: Edward Bace & Ana Ferreira | (2020) Regulation's influence on EU banking efficiency: An evaluation post crisis, Cogent Economics & Finance, 8:1, 1838735, DOI: 10.1080/23322039.2020.1838735

To link to this article: https://doi.org/10.1080/23322039.2020.1838735

© 2020 The Author(s). This open access article is distributed under a Creative Commons Attribution (CC-BY) 4.0 license.



6

Published online: 27 Oct 2020.

|--|

Submit your article to this journal 🖸

Article views: 1082



View related articles

View Crossmark data 🗹



Citing articles: 5 View citing articles 🗹





Received: 05 July 2020 Accepted: 09 October 2020

*Corresponding author: Edward Bace, Accounting & Finance, Middlesex University Business School, London, UK E-mail: E.Bace@mdx.ac.uk

Reviewing editor: David McMillan, University of Stirling, Stirling, UK

Additional information is available at the end of the article

FINANCIAL ECONOMICS | RESEARCH ARTICL Regulation's influence on EU banking efficiency: An evaluation post crisis

Edward Bace¹* and Ana Ferreira¹

Abstract: This paper examines the impact of regulatory policies on banking market efficiency using a sample of 678 commercial banks from 21 European Union countries for the post-crisis year 2010, controlling for bank-specific and country-specific variables. Data on regulation, supervision and monitoring variables, and activity restrictions are from the most recent Bank Regulation and Supervision Survey database conducted by the World Bank, published 2012. Besides these we incorporate bank size, equity, market share, government ownership, and growth of Gross Domestic Product per capita, employing an Ordinary Least Squares method. Focus is on two alternative measures of banking market efficiency: net interest margin and overhead costs (operating expenses to assets). Elevated levels of these two ratios should indicate a low level of banking efficiency. The evidence suggests that the link between capital regulation and banking efficiency is not robust enough to control for other regulatory variables. Results confirm that activity restrictions have a negative and significant impact on banking efficiency. Policies encouraging official supervisory power do not enhance efficiency of the banking sector. The only approach positively and statistically significantly associated with efficiency is private monitoring. This leads to the suggestion that government regulation and supervision should be more focused on promoting transparency of information.

Subjects: Economics; Finance; Business, Management and Accounting

Keywords: banking; efficiency; regulation; European Union JEL Classifications: G14; G21; G28

ABOUT THE AUTHORS

Edward Bace is a senior lecturer in finance at Middlesex University London's Business School. He has an MBA from New York University, and a PhD from The University of Michigan, is a CFA, Chartered MCSI, and a Member of the Professional Risk Managers' International Association (PRMIA). He has been involved with the EBRD, CFA Institute, BTRM, and the Chartered Institute for Securities & Investment.

Ana Ferreira earned a Master's degree in Banking and Finance from Middlesex University, and has a Bachelor's degree in Business Administration from University of Porto, specialising in economics and financial management. She now works in audit and assurance for KPMG Portugal.

PUBLIC INTEREST STATEMENT

This article focuses on the important topic of banking efficiency, which has wide-ranging implications for fairness and effectiveness of banking services for customers and other stakeholders. Key determinants of banking efficiency include the quality of banking regulation and supervision in a given jurisdiction, as well as the scrutiny of market participants and investors (private monitoring) in relation to the structure of a banking system. The study presented here concludes that, for banks in the European Union, the most meaningful influence on efficiency is that of private monitoring, which has implications for regulatory and government policy in promoting efficiency of a banking system.





 \odot 2020 The Author(s). This open access article is distributed under a Creative Commons Attribution (CC-BY) 4.0 license.

1. Introduction

The global financial crisis has revived the importance of the banking system to the economy. As a result of the crisis, bank regulations have become more stringent to prevent future bank failures and safeguard the economy from negative externalities. In this regard, policymakers concentrate on which regulation is the best to achieve these goals. Thus, different regulatory approaches should be analysed as well as the effects of combining approaches. The most common are capital regulation, activity restrictions, official supervisory power and private monitoring. Following this, the present research studies each of these regulatory measures.

The globalization of banking services, the increasing competition and the technological changes have emphasized the importance of banking efficiency. Since banks act as financial intermediaries between lenders and borrowers, banking efficiency plays a crucial role in the social welfare. It is important that banks work to minimize costs, thus promoting efficient utilization of resources. Therefore, enhancing the efficiency of the banking system is vital to achieve greater social welfare. We rely on two accounting ratios to measure banking efficiency: net interest margin and overhead costs. While imperfect, these measures can signal intermediation inefficiency and excessive market power. Hence, higher net interest margin and higher overhead costs reflect lower levels of banking efficiency.

Nevertheless, banking literature provides conflicting predictions about the impact of bank regulation on banking efficiency. As argued by Barth et al. (2006), the different arguments usually reflect the two divergent views, which are the public interest view and the private interest view. The public interest view supports that government operates to enhance social welfare. The purpose of regulating the banking sector is to promote banking efficiency and prevent bank failure. Conversely, the private interest view states that government uses regulatory policies to satisfy political interests (Shleifer & Vishny, 1998). In line with this argument, regulation may exert adverse effects on banking efficiency.

The most common justification for any regulation stems from market failures due to asymmetries of information, externalities and market power. However, there is still no consensus concerning whether the banks should be regulated, and which regulatory policy generates better results. Some argue that capital regulation decreases the occurrence of bank runs, cushioning the economy from major losses and reduce agency problems between bank owners and depositors, especially through deposit insurance. Others argue that stringent capital requirements encourage risk-taking behaviour (Blum, 1999; Boyd et al., 1998; Furlong & Keeley, 1989). By allowing banks to engage in broader activities, banks may become so important to the economy that they are deemed to be "too big to fail", thus difficult to discipline. The power enjoyed by such complex banks may impede competition in the banking sector. On the other hand, restricting activities impedes the exploitation of economies of scope and scale (Laeven & Levine, 2007). The official supervisory approach stresses that supervisors have the incentives to ameliorate market failures as a consequence of informational asymmetries. Conversely, proponents of private monitoring claim that official supervisors use their power to benefit themselves. Hence, government regulation that promotes private monitoring in terms of more transparency of information may be the most efficacious approach to enhance banking efficiency.

This study assesses empirically the impact of these regulatory measures on banking efficiency. The sample consists of 678 commercial banks from 21 European Union (EU) countries for the year 2010. The information for the regulatory variables is collected from Bank Regulation and Supervision Survey database conducted by the World Bank (Barth et al., 2004, 2008, 2013a, 2006, 2013b; Cihak et al., 2012), which provides unique information on how banks are regulated around the world. From the four surveys available, we use the data from the last survey, which is the first examination of this kind conducted after the financial crisis. It covers information on banking policies in 125 countries and was completed in 2012. We also follow the indexes aggregating the responses from the survey of Barth et al. (2006), (2013a).

This study empirically answers the following research question: *What is the impact of bank regulatory policies on banking efficiency?* In addition to the regulatory variables, bank and country-specific control variables are included in our analysis. We incorporate bank size, bank equity, bank market share, government-owned banks and growth of Gross Domestic Product (GDP) per capita. To estimate the empirical models, we employ the Ordinary Least Squares (OLS) method.

The study is organised as follows. Section 2 summarises the relevant banking literature and the main findings along with the hypotheses we aim to test. Section 3 presents the data and methodology, which includes data sources, measurement of the variables, descriptive statistics, and empirical model. Section 4 presents and discusses the empirical results. Section 5 concludes. The last section also discusses limitations and recommendations for future research.

2. Literature review

In this section, we summarize the prior literature related to the effects of Basel II Capital Accord regulations namely, capital requirements (Pillar 1), official supervisory power (Pillar 2) and market discipline (Pillar 3), in addition to activity restrictions on banking efficiency as well as on bank development, soundness, efficiency and risk incentives. Existing theories regarding different regulatory approaches are examined first, followed by the empirical evidence supporting these theories and hypotheses we aim to test.

Theoretical review

2.1. Capital regulation

There is extensive literature focusing on capital regulations emphasising its importance as a key approach to bank regulation. The capital adequacy requirements stipulate the amount of capital that banks need to hold for a certain level and quality of assets. Consequently, it influences the banking-sector efficiency. In the case of bank owners being required to increase the amount of capital at risk, the gains from increased risk would be counterbalanced by the greater likelihood of capital loss. Hence, capital regulation tends to reduce agency problems between bank owners and depositors (Barth et al., 2006; Kaufman, 1992; Keeley & Furlong, 1990). Moreover, two of the primary functions of capital are risk diversification and lower risk-taking. The former considers capital a buffer which protects the assets from being sold or written off at a very low price. An appropriate level of capital safeguards the debt holders and equity holders. The latter implies that, under more stringent capital regulations, bank managers and owners engage in less risky activities (Chortareas et al., 2012). Based on these arguments, capital regulation is expected to enhance banking efficiency.

However, the theory provides conflicting predictions about the positive effects of capital requirements (Santos, 2001). For instance, Barth et al. (2013b) claimed that the arguments in favour are based on the public interest view and the inherent costs of regulating bank capital as higher barriers and higher taxes charged by governments are not considered. The private interest view opposes capital regulation except when the benefits compensate the costs. Furthermore, some authors find that capital regulation increases risk-taking behaviour (e.g. Besanko & Kanatas, 1996; Koehn & Santomero, 1980). According to Blum (1999), stringent capital requirements increase risktaking behaviour and risk of failure as a result of two effects. First, it lowers profits. If the bank has low profits, the cost of bankruptcy is less costly than the cost of excessive risk-taking. Lastly, it affects the marginal return on risk. In the model presented by the author, the marginal return on risk may increase, and thus the first effect may strengthen, further increasing the risk. Therefore, capital regulation may negatively influence banking efficiency.

Additionally, Pasiouras et al. (2009) highlighted the impact of capital requirements on the quantity and quality of loans, the allocation of asset portfolios and the source of funds. The introduction of regulatory capital requirements on an unregulated banking system may decrease aggregate lending while there is no consensus concerning loan quality (Deli & Hasan, 2017; Kopecky & VanHoose, 2006). Bank managers may also look for other forms of assets rather than

loans, affecting banking efficiency as different assets have different returns and require different management. Despite the advantages of portfolio diversification, little is known about the capacity of banks to manage different assets (VanHoose, 2007). Lastly, capital regulation may influence the decision of banks with respect to the trade-off between debt and equity.

2.1.1. Empirical review

Barth et al. (2004) analysed the effects of bank regulation and supervision on bank development, stability, and performance for 107 countries based on the first survey from 1999 of the Bank Regulation and Supervision Survey database carried out by World Bank. The results indicated that tightening capital requirements lowers the number of non-performing loans (used as a proxy for bank stability) while capital stringency is not robustly linked with banking-sector development and performance as measured by accounting ratios (net interest margin and overhead costs) when controlling for other regulatory and supervisory practices. Likewise, Beck et al. (2006b) addressed the influence of capital regulation on banking system stability using the same data for regulatory measures and found no statistically significant link between the two variables. However, the link is robust when using the capital regulatory index as a control variable for the relationship between bank concentration and fragility.

More recently, Barth et al. (2013b) investigated the impact of regulation, supervision, and monitoring on banking efficiency using the non-parametric Data Envelopment Analysis (DEA) approach for a sample of 4050 banks over 1999–2007. The regulatory and supervision variables are from the first three World Bank surveys of Barth et al. (2004), (2006), 2008). They found that stringent capital requirements are significantly associated with higher levels of banking efficiency and also the possible existence of a connection between banking efficiency and safety. Similarly, Chortareas et al. (2012) provided evidence of a positive and significant relationship between capital regulation and banking efficiency using the DEA approach¹ as well as using accounting ratios: net interest margin and cost-toincome ratio. The study comprised 22 EU countries for the years 2000–2008 using the information on regulatory variables from the World Bank dataset. These findings are consistent with the view that strengthening capital regulation is associated with higher levels of banking efficiency.

Previous studies (Demirguc-Kunt et al., 2008) on the soundness of the banking system have highlighted the importance of regulation and supervision for sounder banks. For instance, Pasiouras et al. (2006) contributed to this area researching the relationship between the Fitch individual bank ratings and bank regulations (World Bank database) with a sample of 857 banks for the year 2004. Their results revealed that lower ratings are attributed to banks operating in markets with tighter capital regulation. Therefore, stringent capital regulation is negatively associated with bank soundness. This is consistent with the argument that banks in countries that promote tighter capital adequacy requirements take on greater risk but inconsistent with the positive influence that capital regulation exerts on bank efficiency.

Regarding risk-taking, Laeven and Levine (2009) conducted a study of 279 banks in 48 countries analysing the risk-taking behaviour by banks, ownership structures and bank regulations. The empirical evidence of their study has shown that the interaction between capital regulations, assessed by minimum capital requirements and capital stringency index, and risk critically depends on the ownership structure of each bank. Thus, while capital regulation decreases risktaking in widely held banks, it has an adverse effect when the bank has a large owner. This finding confirms that capital stringency reduces the benefits of owning a bank, hence large and powerful owners undertake riskier activities in response to more stringent capital regulations. Moreover, the paper emphasized the importance of including ownership structure in the analysis of the effects of capital regulations on risk-taking behaviour.

In line with previously mentioned studies on banking efficiency and considering the empirical evidence, we aim to test the following hypothesis:

Hypothesis 1: Capital regulation does not have a positive or significant impact on banking efficiency.

2.2. Activity restrictions

Although activity restrictions are not a pillar of Basel II, they have been stressed as fundamental to the study of the impact of regulation on the proper functioning of the banking sector. The question is whether the existence of financial conglomerates² is beneficial to the performance of the banking system. Nevertheless, there is no consensus in theory about the effects of banks engaging in broader activities or restricting their activities.

The information acquired by banks during the lending process may help banks to be more efficient in other activities such as brokering, real estate investment or dealing mutual fund securities. In the same way, by engaging in broader activities, banks acquire information that can be used to improve the process of making loans. Therefore, restricting activities may impede banks from enjoying economies of scale and scope as it limits the exchange of information (Diamond, 1991; Rajan, 1992; Stein, 2002). Moreover, it can lower the franchise value of a bank due to greater difficulty in diversifying revenue streams and thereby narrowing incentives for more efficient conduct. By being responsible for licensing banks and specifying the permitted activities, regulators become more powerful which gives them more opportunities to pursue economic gains (Djankov et al., 2002). Finally, Beck et al. (2006b) presented empirical evidence that tightening activities boosts bank fragility by impeding banks to engage in non-traditional activities is likely to improve banking efficiency.

On the other hand, some (Aron, 1988) argue that expanding the range of financial services may intensify agency problems between owners and shareholders which reduces the market value. Bank managers may engage in broader activities simply because it is more advantageous to them even if it means a lower market value for the financial institution. Therefore, the decision to restrict activities by managers may be due to what benefits them instead of what benefits the firm (Jensen, 1986). Also, moral hazard problems may arise which stimulates risk-taking behaviour (Boyd et al., 1998). Furthermore, it is hard to implement efficient managerial contracts in financial conglomerates as well as handle agency problems (Rotemberg & Saloner, 1994). Another argument is that banks providing a wide range of services may become so important to the economy that they are deemed to be "too big to fail" thus, difficult to discipline. Under these circumstances, activity restrictions may positively affect banking efficiency. According to Laeven and Levine (2007), diversification of activities decreases the market value of the financial conglomerate since the benefits of enjoying economies of scope are not significant enough to offset the negative effects of leading with a variety of activities. Consequently, restricting bank activities may enhance banking efficiency.

Researchers (King and Levine, 1993) focusing on the impact of activity restrictions on banking efficiency and development have concluded mixed results for the association between the two variables. The results are consistent with the view that relaxing restrictions on bank activities improves banking efficiency by increasing opportunities for diversifying sources of income. Barth et al. (2004) found that while regulation of activities negatively affects bank development, there is no robust link between this regulatory measure and net interest margin or overhead costs or non-performing loans. The former finding is notably important since the development of the banking system directly influences economic growth.³ In addition, their study supported that a banking system with greater freedom on activities is less likely to suffer a major crisis. Kim et al. (Role of financial regulation and innovation in the financial crisis, Kim et al., 2013) carried out a study on the role of regulatory measures in the financial crisis⁴ using the surveys on bank regulation for 143 countries. Unlike Barth et al. (2004), they concluded that activity restrictions reduce the likelihood of a banking crisis. Therefore, diversification drives banks to engage in riskier activities, further increasing bank fragility.

Chortareas et al. (2012) confirmed that greater restrictions on bank activities exert a positive impact on net interest margin, cost-to-income ratio and inefficiency, which means an overall

negative impact on banking efficiency. Barth et al. (2013b) also corroborated that stricter restrictions on bank activities are negatively associated with the efficiency of the banking system, using the DEA approach. Moreover, Demirguc-Kunt et al. (2004) investigated the influence of bank regulations on the cost of financial intermediation using as proxies the net interest margin and the overhead expenditures. The sample comprised 1400 commercial banks in 72 countries, from 1995 through 1999. Similar to the studies mentioned above, the regulatory measures were obtained from the World Bank surveys. The results showed that greater activity restrictions are significantly associated with greater net interest margin, consequently greater operational inefficiency and market power. With respect to the soundness of the banking system, Pasiouras et al. (2006) found that lower ratings are allocated to banks operating in markets with stricter activity regulations. Thus, restricting bank activities negatively influence banking efficiency.

Laeven and Levine (2009) proposed in their study that tighter restrictions on bank activities increase risk-taking behaviour when the bank has a sufficiently large owner. If the bank is widely held, the effects of restricting activities on risk are not significant. Since regulation on activities lowers the profits, bank owners may engage in riskier activities to compensate for losses. In this respect, Gonzalez (2005) evaluated the impact of activity restrictions on risk incentives in a study with 251 banks in 36 countries from 1995 to 1999. Nevertheless, the evidence has confirmed that greater regulatory restrictions are related to fewer risk-taking incentives after isolating the relationship between bank charter value⁵ and risk-taking incentives and between regulatory restrictions and bank charter value. The findings support that greater freedom on bank activities promotes risk-taking as it provides banks more opportunities to assume greater risk. Hence, the two studies provided conflicting predictions about the effects of activity restrictions on bank risk.

Considering the empirical evidence presented in previous banking literature on the effects of activity restrictions on banking efficiency, we aim to test the following hypothesis:

Hypothesis 2: Activity restrictions have a negative and significant impact on banking efficiency.

2.3. Official supervisory power

The official supervisory power concerns the degree to which bank supervisors can take actions against bank owners, bank managers and bank auditors in different circumstances. While there is an extensive literature on the importance of empowering official supervision of banks, empirical evidence (Levine, 2005) shows no consensus about the effects on bank performance. Since the costs of supervising banks are excessive, there are few incentives to monitor banks. Proponents of public interest view claim that supervisors have incentives to ameliorate market failures through the monitoring and discipline of banks (Beck et al., 2006a), thus undermining corruption and strengthening the intermediary function of banks. Moreover, the existence of informational asymmetries makes banks vulnerable to contagion risk and possible bank failure. Powerful supervisors can intervene effectively under these circumstances. Besides, with the adoption of deposit insurance, banks may undertake greater risk and depositors may have less incentives to supervise banks. Official supervisors may be capable of preventing such situations (Barth et al., 2004; Beck et al., 2006b). Given the previous arguments, it is expected that official supervisory power enhances banking efficiency.

On the contrary, proponents of private interest view claim that official supervisors use their power for their own interest. As powerful supervisors have access to insider information, they may use it for private and political benefits (Beck et al., 2006b). Political candidates may bribe supervisors to extract votes and donations for their campaign (Djankov et al., 2002; Quintyn & Taylor, 2003). Hence, official supervisory power may be associated with higher levels of corruption, negatively influencing banking efficiency. Additionally, supervisors may pursue private interests instead of public interests to enhance their reputation as being able to monitor banks, especially when there are uncertainties about their capacity to do so (Boot & Thakor, 1993). Thus, powerful

supervisors may adversely affect banking operations. Therefore, official supervisory power may positively or negatively affect the corporate governance of banks, the lending process (by encouraging corruption), and the intermediary function of banks.

Nevertheless, the argument that an independent supervisory entity has positive influence on banking efficiency is consensual among the two opposing views. Being independent, the supervisors are less likely to exert power on banks to serve political and private interests. Thus, independent supervisors could monitor the corporate governance of banks more professionally. Moreover, they may better guide banks by giving them advice on how to improve (Barth et al., 2013b).

The empirical evidence indicates that official supervisory power is not significant in explaining bank performance itself. Barth et al. (2004) found that empowering official supervision of banks is not significantly associated with bank development, bank performance or the number of non-performing loans. The link between supervision and the likelihood of a banking crisis also proved to be non-significant. Similarly, the regressions in the study of Barth et al. (2013b) pointed to an insignificant relationship between official supervisory power and banking efficiency. The study also evaluated the impact of independent supervisory authorities and the interaction between the two variables. Their results suggested that while official supervisory power does not enhance banking efficiency itself, promoting the independence of supervision may strengthen banking efficiency. Furthermore, the interaction term confirmed that empowering official supervisors. This is consistent with the argument that independent supervisors are less likely to pursue political and private interests rather than social interests, improving the corporate governance of banks.

Levine (2005) assessed the impact of supervision, specifically the official supervisory approach and the private monitoring approach, on banking efficiency measured by overhead costs using the data for the supervisory measures from Barth et al. (2004). The connection between official supervision and overhead costs has proven to be statistically insignificant. The research emphasized that the aim of the official supervisory approach may be to lower bank fragility instead of enhancing bank performance as a possible justification. Notwithstanding the main finding, the data indicated that empowering official supervision intensifies corruption in bank lending. Hence, the data is consistent with the private interest view.

Different results are suggested by Chortareas et al. (2012). The evidence indicated that official supervisory power has a positive impact on banking efficiency, which confirms the predictions of the public interest view. The results held using inefficiency, net interest margin or cost-to-income ratio as independent variables, with coefficients being negatively significant in all regressions. However, when controlling for private monitoring, the impact turned negative. One explanation may be that strengthening official oversight in less developed countries leads to increased government participation and thus undermines the reliability of the lending process.

Beck et al. (2006a) evaluated the association between supervision and corruption in bank lending in a study with 2500 firms from 37 countries using the information on supervision from Barth et al. (2008; 2004; 2006). Corruption in bank lending is proxied by the extent to which bank supervisors' corruption is a barrier for firms to get external finance. The results supported that strengthening official supervisory power is positively and significantly related to higher levels of corruption. This suggests that powerful supervisors do not improve social welfare, they improve their own welfare. Therefore, policies that encourage official oversight to enhance banking efficiency may have opposite outcomes, ultimately increasing corruption in the banking system.

The analysis of Pasiouras et al. (2006) concluded that more powerful supervision is adversely associated with bank ratings. Thereby, countries that promote official supervision to take specific actions on bank managers and owners obtain lower bank ratings. Under these circumstances, official supervisory power is negatively related to the soundness and efficiency of the banking

system and positively related to bank corruption. The study of Demirguc-Kunt et al. (2008) which examined the influence of compliance with the Basel Core Principles on Moody's ratings noted that countries with a higher degree of compliance are better rated, although empowering official supervision per se does not influence bank ratings.

The evidence presented in the literature on the effects of official supervisory approach on banking efficiency is divided into two viewpoints, those that found negative and significant effects and those that found insignificant effects. This leads us to the following hypothesis:

Hypothesis 3: Official supervisory power has a negative impact on banking efficiency.

2.4. Private monitoring

Private monitoring refers to the extent to which banking regulations require banks to disclose accurate and detailed information to the public and improve contract enforcement mechanisms. It is believed that bank supervision should focus on reducing barriers such as the costs of producing and disclosing information. Thus, private investors could monitor more efficiently, strengthening the corporate governance of banks (Hay & Shleifer, 1998). Proponents of the private monitoring approach often express doubts about the effectiveness of the official supervisory approach. Therefore, the arguments in favour of the former approach tend also to be critiques to the latter approach and vice versa (Levine, 2005).

First, as supervisors are not usually bank shareholders, their incentives may conflict with incentives from private investors regarding bank discipline, consequently agency problems may arise (Barth et al., 2013b). Second, supervisors may be driven by self-interests. Politicians and banks may influence official supervisors to act in their own interests by enriching insiders and channelling credit to some specific companies according to the private interest view. Hence, empowering private monitoring over supervisory power may prevent corruption in bank lending from intensifying, improving banking operations at the same time (Beck et al., 2006a). Third, remuneration of supervisors is low in some countries which incentivises them to move to the banking sector. These circumstances may create different motivations for supervisors regarding bank monitoring (Barth et al., 2004). Lastly, while recognizing the existence of market failures that interferes with the effectiveness of private monitoring, Shleifer and Vishny (1998) emphasized that official oversight may have a more significant negative impact on banks due to the great scale of government failures. Following this logic, the benefits of overcoming market failures may not produce sufficient positive effects to promote official supervisory power over private monitoring.

Nevertheless, there are arguments against empowering private monitoring. Private monitoring may not enhance bank performance in countries with less developed legal systems, poor capital markets and ineffective financial accounting policies as people in these countries do not trust in private monitoring. Therefore, they may not be able to benefit from an efficient monitoring with the private sector, but they may benefit from official oversight. Since the banking system is very complex and opaque, even the private sector in more developed countries may find it hard to monitor banks efficiently (Barth et al., 2004). From this standpoint, strengthening private monitoring may lead to bank customers being misled, and consequently deteriorating banking efficiency. Furthermore, costs related to disclosure of accurate information to the public such as enforcement costs, costs to prepare formal disclosure documents and costs of divulging delicate information to competitors are excessive, thus it negatively impacts banking efficiency (Duarte et al., 2008).

While the arguments mentioned above have supported one supervisory approach over the other between official supervision and private monitoring, the two approaches are not mutually exclusive as stated by Levine (2005). Indeed, countries are able to adopt both approaches, that is,

requiring banks to disclose precise information to the public at the same time that official supervisory agencies directly monitor and discipline banks.

Most of the existing studies have concluded that empowering private monitoring in terms of more financial transparency has a good impact on the banking system. For instance, Barth et al. (2004) observed that private monitoring positively influences bank development and negatively influences the amount of nonperforming loans and net interest margin. The study, however, failed to find a strong impact on bank crises when controlling for other variables. Barth et al. (2013b) found that private monitoring exerts positive effects on banking efficiency, specifically they found that strength of external auditor, bank accounting informative and certified audit requirements are related with higher levels of banking efficiency. Similar results were obtained in the study of Levine (2005) which concluded that empowering private monitoring reduces the overhead costs, boosting banking efficiency. One particularity of this study was that the effect lost significance in the regression that included GDP per capita. This means that GDP per capita is highly correlated with the degree of information disclosure. In a broader perspective, the economic development of a country is influenced by the implemented supervisory approach, just as the effectiveness of bank supervision depends on the economic development of the country.

Conversely, Chortareas et al. (2012) provided evidence of a negative relationship between private monitoring and banking efficiency. The results should be interpreted carefully since they used a sub-sample of countries for regressions that included this variable, excluding economically large countries as UK, Germany, France and Spain. A possible reason for this finding is that producing and disclosing information is costly. Hence, banking efficiency may be adversely affected.

Beck et al. (2006a) noticed that empowering private monitoring reduces corruption in bank lending. Their paper also established a positive association between bank lending integrity indicators and the private monitoring approach. Hence, corruption of bank officials is less a barrier for companies to raise capital in countries that promote information provision and countries with developed legal systems and efficient governments. The findings are consistent with the argument that private monitoring reduces bank corruption, enhancing banking efficiency.

Pasiouras et al. (2006) analysed the impact of accounting and disclosure requirements and auditing requirements on bank credit ratings and found that both coefficients appear negative in the regressions. This is inconsistent with studies that indicate that empowering private monitoring enhances bank performance as well as with those that indicate that accounting and auditing systems reduce bank risk (Fernandez & Gonzalez, 2005). Nevertheless, the coefficients are only significant at 10% level, which is consistent with the findings of Barth et al. (2004) regarding the impact of private monitoring on the likelihood of a banking crisis. Additionally, the empirical evidence of Demirguc-Kunt et al. (B2008) supported that compliance with dissemination of accurate information is positively associated with bank soundness. Thus, banks in countries that strengthen private monitoring are better rated.

Focusing on the findings of the aforementioned studies about the effects of strengthening private monitoring on banking efficiency, we aim to test the following hypothesis:

Hypothesis 4: Private monitoring does not have a positive or significant impact on banking efficiency.

This study investigates the impact of bank regulatory measures namely, capital regulation, activity restrictions, official supervisory power and private monitoring on banking efficiency in the EU. The methodology we follow is closely related to various research on banking efficiency. To measure banking efficiency, we use two accounting ratios: net interest margin and overhead costs. Likewise,

Barth et al. (2004), Demirguc-Kunt et al. (2004), and Levine (2005) used accounting ratios as opposed to studies that used efficient frontier analysis (Barth et al., 2013b). Chortareas et al. (2012) measured banking efficiency using accounting ratios as well as the DEA technique. Both methods produced the same results. Therefore, we decide to use accounting ratios. However, it is important to emphasize that bank net interest margin and overhead costs are imperfect measures of banking efficiency. In order to overcome this limitation, we control for bank-specific and country-specific variables.

3. Data and methodology

In this section, we present information about the data and the methodology employed in our research. We discuss the data collected, sources and sample selection, the definition of the variables and the descriptive statistics. This is followed by the introduction of the econometric models, description of the estimation procedure and the tests performed to obtain accurate results.

3.1. Data

This study aims to evaluate the impact of bank regulatory measures on banking efficiency. To assemble the cross-section data, we include commercial banks operating in 21 EU countries, namely: Austria, Belgium, Bulgaria, Croatia, Cyprus, Denmark, Finland, France, Greece, Hungary, Ireland, Italy, Luxembourg, Malta, Netherlands, Poland, Portugal, Romania, Slovakia, Slovenia and Spain.

The dataset is compiled from two primary sources, SNL Financial database provided by S&P Global Market Intelligence and World Bank databases. The bank-specific variables: net interest margin, overhead costs, bank size, bank equity and bank market share are obtained from SNL Financial database. The annual growth rate of GDP per capita is obtained from the World Bank World Development Indicators (World Bank, 2010). The regulatory measures: capital regulation, activity restrictions, official supervisory power and private monitoring, as well as the government-owned banks, are obtained from the last survey of the Bank Regulation and Supervision Survey carried out by the World Bank (Barth et al., 2004, 2008, 2013a, 2006, 2013b; Cihak et al., 2012). As the answers from this survey correspond to the year 2010, we collect data for the same year. Moreover, we follow the indexes constructed and made available by Barth et al. (2006) and (2013a), which aggregates the responses to individual questions into indexes.

Since the availability of data varies by country and by bank, we edit the raw data with the purpose of minimising the errors in our analysis. Therefore, our sample comprises only the observations for which all variables are available. The initial sample included 28 EU countries.⁵ The Czech Republic and Sweden did not respond to the survey. Hence, we do not have information on any regulatory variable for these countries. Therefore, they are dropped from our sample, resulting in 1218 banks from 26 countries. After dropping the observations without information for the bank-specific variables, there are 1019 banks from 24 countries, Latvia and Lithuania are excluded. Although the United Kingdom, Germany and Estonia responded to the survey, there are missing values for the regulatory variables. For the United Kingdom, the information is not available for official supervisory power; for Germany, information is not available for activity restrictions; for Estonia, information is not available for private monitoring. Consequently, the final sample used in our study to examine the impact of regulation on banking efficiency consists of 678 banks from 21 EU countries.

3.1.1. Net interest margin and overhead costs

To measure banking efficiency, we use two accounting ratios: net interest margin and overhead costs. We collected data for net interest margin and overhead costs from 2010 since the information for bank regulation is from this year. Net interest margin equals interest income minus interest expense divided by total assets. The net interest margin measures the gap between what banks pay to their lenders and what banks receive from their borrowers, hence it reflects the conventional lending and borrowing operations of the bank. Similar to previous papers, we interpret the net interest margin as a signal of intermediation inefficiency and excessive market power, allowing banks to charge higher interest. Thus, higher values of net interest margin reflect lower levels of banking efficiency.

Overhead costs equal operating expenses divided by total assets. High overhead costs reflect unwarranted managerial perquisites as well as higher market power, which is inconsistent with the efficient intermediation of the banking system. Hence, higher values of overhead costs are associated with lower levels of banking efficiency. It is expected that both dependent variables produce identical results (Barth et al., 2008; Demirguc-Kunt et al., 2004; Levine, 2005).

In our econometric model, we use the logarithm values of both net interest margin (LOGNIM) and overhead costs (LOGOC) with the aim of normalising the distribution of the variables and consequently decrease their fluctuation. Graphically, we can perceive that the distribution of the variables before the logarithm values exhibits high skewness and high kurtosis. Using the logarithm, the values of skewness and kurtosis are closer to the values for the standard normal distribution.

Differences across banks in net interest margin and overhead costs may reflect differences in bank activities, business systems, or asset allocations rather than differences in banking efficiency. In order to mitigate possible interpretational issues with the measures of banking efficiency and be able to find the independent relationship between them and bank regulation, we control for the bank and country-specific variables.

3.1.2. Regulatory variables

With the purpose of evaluating the impact of bank regulation on banking efficiency, we include four regulatory measures in our study, namely, capital regulation, activity restrictions, official supervisory power, and private monitoring. We provide the description of these variables and the range of the corresponding indexes constructed by Barth et al. (2006), (2013a)). The regulatory measures data were collected from the last survey. Hence, it refers to the year 2010.

The capital regulatory variable (CAPREG) is an index that incorporates the initial capital stringency and the overall capital stringency. The initial capital stringency concerns whether the sources of regulatory capital can include assets other than cash, government securities or borrowed funds and whether the regulatory/supervisory authorities verify them. The overall capital stringency concerns whether Basel I is the applied capital adequacy regime, whether the capital requirement reflects market or credit risk, whether market value losses are deducted, and the fraction of revaluation gains permitted. The capital regulatory index ranges from 0 to 10 with higher values indicating greater capital stringency.

The activity restrictions variable (ACTRES) is an index that measures the degree to which banks are allowed to engage in non-traditional activities, specifically, securities, insurance and real estate activities. Securities activities include brokering, underwriting, dealing, and all aspects of the mutual fund industry. Insurance activities include underwriting and selling. Real estate activities include management, investment and development. The activity restrictions index ranges from 4 to 16 with higher values indicating greater restrictiveness.

The official supervisory power variable (SPOWER) is an index that concerns whether official supervisors have the power to act on bank owners, auditors, and managers in different circumstances with the intention of preventing and correcting problems. This includes whether the supervisory authority is allowed to share information with external auditors, declare bank insolvency, supersede the rights of shareholders, change bank managers and directors, build a legal case against auditors in case of negligence, require banks to constitute provisions to cover losses, require banks to reduce or suspend dividends, bonuses and other remunerations and whether auditors are required to report unlawful activities. The official supervisory power index ranges from 0 to 14 with higher values indicating a greater power of supervisory authorities.

The private monitoring variable (PMONIT) is an index that measures the degree to which supervisors require banks to disclose accurate and detailed information to officials and the public, promoting the monitoring of banks by the private sector. It involves whether professional external auditors audit

banks, whether international and domestic rating entities rate banks, whether there is an explicit deposit insurance scheme and the compensation received by insured depositors the last time a bank failed, whether banks are required to produce consolidated accounts and disclose off-balance sheet items, whether accrued or unpaid interest or principal are included in the income statement on performing and non-performing loans, whether bank directors are legal responsible for the accuracy of the information disclosed and whether subordinated debt is allowed. The private monitoring index ranges from 0 to 12 with higher values indicating greater private monitoring of banks.

3.1.3. Control variables

At the bank-level, we use three key control variables: bank size, bank equity and bank market share. At the country-level, we control for the GDP growth and the government-owned banks. We use 2010 values as the data for the regulatory variables corresponds to this year. Since we find the same results when we omit the control variables, endogeneity problems are not biasing our conclusions regarding the effects of bank regulation on banking efficiency.

Bank size (LOGTA) is measured as the logarithm of total bank assets in US dollars. We control for bank size to capture the effect on bank lending behaviour. Moreover, this variable may have a considerable impact on net interest margin and overhead costs in case of increasing returns to scale in the banking industry. Bank equity (EQTA) is measured as the book value of bank equity divided by total bank assets. We use this variable as previous papers (e.g. Berger, 1995) supported that well-capitalised banks face lower bankruptcy costs, hence lower funding costs. Therefore, higher values of bank equity may imply higher net interest margin and overhead costs. Bank market share (MS) equals the total assets of the bank divided by the total commercial bank assets in the country. Banks with greater market share may experience greater market power, consequently greater net interest margin and overhead costs.

Government-owned banks (GOVERN) equals the fraction of the banking system's assets that is held by banks with 50% or more government-ownership. This variable measures the ownership structure of the banking system. La Porta et al. (2002) suggested that government-owned banks tend to allocate resources to benefit politicians and therefore tend to be less efficient. According to this perspective, we expect government-owned banks to exert a positive impact on net interest margin and overhead costs. GDP growth (GDPGR) equals the annual growth rate of GDP per capita. We include this variable to control for business-cycle forces of the macroeconomic environment. Since the demand for financial services depends on the cyclical conditions, GDP growth may have a positive influence on banking efficiency.

3.1.4. Descriptive statistics

This paragraph presents and discusses the key descriptive statistics for the dependent and independent variables used in our regression analysis.

Table 1 shows the summary statistics of all the variables used to study the impact of bank regulatory measures on banking efficiency. The average for the LOGNIM is -4.24, with a minimum of -11.10 and a maximum of -1.58. The number of observations of this variable (671) is lower compared to the number of observations of all the other variables (678). The difference refers to the banks from our sample with values of net interest margin equal to 0, since the logarithm of 0 is not computable. The average for the LOGOC is -4.09, with a minimum of 9.95 and a maximum of -0.42. Regarding the regulatory variables, CAPREG has an average of 6.70 with a standard deviation of 1.56, a minimum of 4 and a maximum of 9. As the range of this index is from 0 to 10, we can conclude that the 21 EU countries have a medium-high level of capital stringency. For the ACTRES, the average is 6.21 with a standard deviation of 1.7, a minimum of 4 and a maximum of 11. The banks in our sample are less regulated in terms of engaging in securities, insurance, and real estate activities, compared to the capital as the range of the former is from 4 to 16. The average SPOWER is 11.29 with a standard deviation of 1.9, a minimum of 5 and a maximum of 9. Since this index ranges from 0 to 14, we can conclude that the official supervisory authorities in the considered countries have great

Table 1. Descriptive statistics						
Variable	Mean	Std. Dev.	Minimum	Maximum	Observations	
LOGNIM LOGOC	-4.24	0.98	-11.10	-1.58	671	
	-4.09	1.24	-9.95	-0.42	678	
CAPREG ACTRES	6.70	1.56	4	9	678	
	6.21	1.70	4	11	678	
SPOWER PMONIT	11.29	1.90	5	14	678	
	8.32	1.06	6	11	678	
LOGTA EQTA MS GOVERN GDPGR	6.53	1.00	3.42	9.51	678	
	0.12	0.13	-0.01	1.00	678	
	0.03	0.07	0.00	0.58	678	
	0.07	0.09	0	0.51	678	
	0.01	0.02	-0.06	0.05	678	

Note: This table presents the mean, standard deviation, minimum, maximum and number of observations of all variables used in this study for a sample of 21 EU countries for the year 2010. Sources: SNL Financial database, World Bank: Bank Regulation and Supervision Survey, and World Bank: World Development Indicators and own calculations. The bank-level variables include logarithm of net interest margin (LOGNIM), logarithm of overhead costs (LOGOC), bank size (LOGTA), bank equity (EQTA) and bank market share (MS). The country-level variables include capital regulation (CAPREG), activity restrictions (ACTRES), official supervisory power (SPOWER), private monitoring (PMONIT), government-owned banks (GOVERN) and annual growth rate of GDP per capita (GDPGR).

power to take actions against bank owners, managers and auditors. Lastly, the average for the PMONIT is 8.32 with a standard deviation of 1.06, a minimum of 6 and a maximum of 11. Given that the range of this index is from 0 to 12 and considering the statistics, we can conclude that the banking regulations in the countries of our sample promote private monitoring of banks. The fact that the countries in our sample promote both official supervisory power and private monitoring is consistent with the theory that the two approaches are not mutually exclusive as discussed in the literature review. Considering the control variables, the average LOGTA is 6.53, with a minimum of 3.42 and a maximum of 9.51. The average for the EQTA is 0.12, with a minimum of -0.01 and a maximum of 1. For the MS, the average is 0.03, the minimum is approximately 0, and the maximum is 0.58, meaning that there is at least one bank in a country that holds more than 50% of the total commercial bank assets in that country. The GOVERN has an average of 0.07, a minimum of 0 and a maximum of 0.51. These statistics reflect that there are one or more countries in our sample without government-owned banks, as well as at least one country where government-owned banks hold more than 50% of total banking assets in that country. The average for the GDPGR is 0.01, with a minimum of -0.06 and a maximum of 0.05, reflecting that there are countries with negative growth rates of GDP per capita and even the highest value reflects positive but low growth rate.

3.2. Model description

To assess the effect of bank regulatory measures on banking efficiency while controlling for bankspecific and country-specific variables, we estimate the following two models:

$$LOGOC_{i,t} = \alpha + \beta_1 CAPREG_t + \beta_2 ACTRES_t + \beta_3 SPOWER_t + \beta_4 PMONIT_t$$

$$\beta_5 B_{i,t} + \beta_6 C_t + \varepsilon_i$$
(1)

$$LOGOC_{i,t} = \alpha + \beta_1 CAPREG_t + \beta_2 ACTRES_t + \beta_3 SPOWER_t + \beta_4 PMONIT_t$$

$$\beta_5 B_{i,t} + \beta_6 C_t + \varepsilon_i$$
(2)

Where *i* refers to bank *i*, t refers to country t, α is the intercept, $B_{i,t}$ is a vector of bank-specific variables for bank *i* in country *t*, C_t is a vector of country-specific variables in country t and $\varepsilon_{i,t}$ is the residual. The

measures of banking efficiency are two accounting ratios: the logarithm of net interest margin and logarithm of overhead costs. In model (1), the dependent variable is $LOGNIM_i$ which represents the logarithm of net interest margin for bank *i* in country *t*. In model (2), the dependent variable is $LOGOC_i$, which represents the logarithm of overhead costs for bank *i* in country *t*. *CAPREG*_t is the capital regulation variable in country *t* and θ_1 is the coefficient which explains the effect of capital regulation on banking efficiency. ACTRES_t corresponds to the activity restrictions variable in country *t* and θ_2 is the coefficient describing the impact of activity restrictions on banking efficiency. *SPOWER*_t denotes the official supervisory power variable in country *t* and θ_3 is the effect of official supervisory power on banking efficiency. *PMONIT*_t is the private monitoring variable in country *t* and θ_4 describes the influence of private monitoring on banking efficiency. The vectors *B*_i, and *C*_i including the bank-specific variables and the country-specific variables, respectively, are defined as follows:

 $B_{i,t} = (LOGTA_{i,t}, EQTA_{i,t}, MS_{i,t})$

$C_t = (GOVERB_t, GDPGR_t)$

The vector B_i contains bank size, bank equity and bank market share. Bank size is proxied by the logarithm of total assets for bank *i* in country *t* (*LOGTA*_I). Bank equity is measured as the equity to assets ratio for bank *i* in country *t* (*EQTA*_i). Bank market share is assessed by the ratio between the total assets of the bank and the total commercial bank assets in the country for bank *i* in country *t*. The vector C_t includes the government-owned banks variable in country *t* (*GOVERN*_i) and the annual growth rate of GDP per capita in country *t* (*GDPGR*_t).

Previous to the regression of model 1 and 2, it is necessary to check whether the variables are suitable for our cross-section data and whether the assumptions about our data are fulfilled. Therefore, we examine the graphical distribution and the measures of skewness and kurtosis of the variables. The skewness and kurtosis values for standard normal distribution are 0 and 3, respectively. The only variables whose values are significantly different from the standards are net interest margin and overhead costs. As previously mentioned, we use the logarithm values to solve this problem by normalising the distribution of the variables. From Figures 1 and 2, we can perceive the difference before and after the logarithm values.

In order to regress the models, we use the OLS estimation as the method that best fits our data. Since heteroskedasticity is often a problem in cross-section data, we test both models for the presence of heteroskedasticity using the Breusch–Pagan test. The null hypothesis of the test refers to the presence of homoskedasticity. The p-value for both models equals 0.00, hence, we reject the null hypothesis. Thus, we use the heteroskedasticity-robust standard errors in our regressions.



Figure 1. Histograms of net interest margin and logarithm of net interest margin, *Source: the authors.*

Figure 2. Histograms of overhead costs and logarithm of overhead costs, *Source: the authors.*



4. Empirical results

In this section, we present and discuss the main results of the estimated regressions. We compare our findings with the expected findings included in our hypotheses, which consider the evidence presented in previous banking literature. Explanations for possible inconsistencies are also examined.

Table 2 presents the regressions results of bank regulation on banking efficiency proxied by the logarithm of net interest margin and overhead costs while controlling for bank-specific and country-specific variables. To recall, we interpret the results as higher values of net interest margin and overhead costs are associated with lower levels of banking efficiency.

Table 2. Regression results					
	(1)	(2)			
	LOGNIM	LOGOC			
CAPREG	0.0353	0.0399			
	(0.0304)	(0.0337)			
ACTRES	0.161***	0.175***			
	(0.0266)	(0.0313)			
SPOWER	0.0177	0.0511**			
	(0.0327)	(0.0242)			
PMONIT	-0.119*	-0.171***			
	(0.0612)	(0.0581)			
LOGTA	-0.197***	-0.445***			
	(0.0493)	(0.0648)			
EQTA	0.158	1.154*			
	(0.578)	(0.658)			
MS	2.063***	2.247***			
	(0.671)	(0.619)			
GOVERN	0.224	-1.169***			
	(0.313)	(0.381)			
GDPGR	-14.89***	-9.589***			
	(2.087)	(2.366)			
Constant	-3.283***	-1.669**			
	(0.858)	(0.684)			

(Continued)

Table 2. (Continued)						
	(1)	(2)				
Observations	671	678				
R-squared F	0.130	0.234				
	15.89	20.89				

Note: This table exhibits the estimates from OLS regressions of model 1 and 2, explaining the relationship between bank regulation and banking efficiency. The dependent variable of model 1 is logarithm of net interest margin (LOGNIM) and model 2 is logarithm of overhead costs (LOGOC). The independent variables of both models are: capital regulation (CAPREG), activity restrictions (ACTRES), official supervisory power (SPOWER), private monitoring (PMONIT), bank size (LOGTA), bank equity (EQTA), bank market share (MS), government-owned banks (GOVERN) and annual growth rate of GDP per capita (GDPGR). The symbols *, **, *** indicate the statistical significance of the estimated coefficients at the 10, 5 and 1% level, respectively. The robust standard errors are in parentheses below the estimated coefficients. The number of observations, R-squared and F statistic of each regression are also presented in this Table.

Our results do not suggest a strong relationship between capital regulation and banking efficiency. While capital regulation enters positively in both regressions, the coefficient is not statistically significant. This is inconsistent with the expected positive and significant effect of capital regulation on banking efficiency described in hypothesis 1. The results are also inconsistent with previous studies (Barth et al., 2013b; Chortareas et al., 2012) that found a significantly positive impact on banking efficiency. On the contrary, Barth et al. (2004) reported an insignificant link between capital regulation and net interest margin or overhead costs, which supports the results in Table 2.

The results also indicate that restricting bank activities is associated with lower levels of banking efficiency, as confirmed by the positive and significant coefficients at the 1% level. The size of the coefficients is substantial. That is, a one unit increase in activity restrictions increases net interest margin by 17.5% (100^* ($e^{0.161} - 1$)) and overhead costs by 19.1% (100^* ($e^{0.175} - 1$)). Moreover, our findings support hypothesis 2 that activity restrictions have a negative and significant impact on banking efficiency. Similarly, Demirguc-Kunt et al. (2004), Chortareas et al. (2012), and Barth et al. (2013b) provided evidence that greater activity restrictions negatively affect the efficiency of the banking system.

Table 2 regressions show that there is a negative relationship between official supervisory power and banking efficiency, which supports hypothesis 3. The link is significant at the 5% level in the second regression when considering overhead costs as the dependent variable. For the first regression, official supervisory power is not significant in explaining net interest margin. This is in accordance with previous banking literature (Barth et al., 2004, 2013b; Levine, 2005) that found no robust impact of official supervisory power on banking efficiency as well as with Chortareas et al. (2012) that found a positive and significant impact on overhead costs when including the private monitoring variable.

As for the private monitoring variable, the coefficient enters significantly and negatively in both regressions. For the net interest margin regression, the coefficient is significant only at the 10% level, while for the overhead costs it is significant at the 1% level. The coefficients are considerably large, hence a one unit increase in private monitoring decreases net interest margin by 12.6% (100* $(e^{0.119} - 1))$ and overhead costs by 18.6% (100* $(e^{0.171} - 1))$). The results confirmed that promoting private monitoring enhances banking efficiency as predicted in our hypothesis 4. This finding is in line with previous studies on banking efficiency (Barth et al., 2004, 2013b; Levine, 2005) that presented empirical evidence in favour of empowering private monitoring. On the other hand, our results are in contrast with the study conducted by Chortareas et al. (2012) that found a positive impact of private monitoring on efficiency levels.

Considering the vector of bank-specific control variables, large banks tend to have lower net interest margin and lower overhead costs. Therefore, bank size is positively and significantly at the 1% level associated with banking efficiency. This may be caused by economies of scale in the banking sector. The finding is in accordance with the evidence presented by Altunbas et al. (2007) and Yildirim and Philippatos (2007) for EU countries. The results for bank equity reveal a positive impact on net interest margin and overhead costs. However, the coefficient is only significant in the second regression at the 10% significance level. This finding confirms that well-capitalised banks have more opportunities for higher net interest margin and overhead costs. Bank market share appears to be an essential driver of banking efficiency as confirmed by the positive and significant coefficients at the 1% level in both regressions. This supports the view that banks with more significant market share experience greater market power and consequently, larger net interest margin and overhead costs.

At the country level, we control for government-owned banks and GDP growth. The government-owned banks positively affect net interest margin and negatively affect overhead costs. Nevertheless, the former relationship is not significant while the latter is significant at the 1% level. The banking literature supports that the presence of the government stimulates corruption in the banking industry, hence promoting banking inefficiency. Our results are inconsistent with this view. Additionally, the annual growth rate of GDP per capita enters negatively in both regressions. The estimated coefficients are significant at the 1% level. This finding indicates that expanding markets, as measured by a higher growth rate, tend to have more efficient banking.

Some of our findings are not consistent with previous literature (Altunbas et al., 2007) on the impact of regulatory measures on banking efficiency, especially regarding capital regulation. Nevertheless, our study differs from the previous ones in a few aspects. First, we are considering the EU countries while most of the studies consider a large cross-country sample. Second, we are using the data from the last survey of the Bank Regulation and Supervision Survey carried out by the World Bank. This is the only study evaluating the impact of bank regulation on banking efficiency using the last survey. For instance, Barth et al. (2013a) and Cihak et al. (2012) presented and discussed the responses to the last survey and compared them to earlier surveys. However, the impact on banking efficiency is not assessed. Lastly, this survey is the first survey conducted after the financial crisis. The global financial crisis has drawn the attention of policymakers to the need to reform banking regulation since it has highlighted problems with the banking sector. Hence, discrepancies with previous studies may exist.

5. Conclusion

The global financial crisis has emphasised the importance of regulatory policies for the proper functioning of the banking system. In this regard, our study contributes to the existing literature by assessing the impact of the Basel II Capital Accord regulations in addition to activity restrictions on banking efficiency. Specifically, we examine the impact of capital regulation, activity restrictions, official supervisory power and private monitoring. In order to do so, we develop a set of hypotheses in line with our predictions considering the existent empirical evidence. We collected data for a sample of 678 banks from 21 EU countries: Austria, Belgium,

Bulgaria, Croatia, Cyprus, Denmark, Finland, France, Greece, Hungary, Ireland, Italy, Luxembourg, Malta, Netherlands, Poland, Portugal, Romania, Slovakia, Slovenia and Spain for the year 2010. We used two alternative accounting ratios to measure banking efficiency, namely net interest margin and overhead costs. We used OLS estimation to regress the empirical models. Also, we employed cross-section data techniques to minimise the errors of our analysis.

The results show that strengthening capital regulation is negatively but not significantly associated with banking efficiency. This finding does not imply that capital regulation is irrelevant for banking efficiency. It merely suggests that the link between capital regulation and banking efficiency is not sufficiently robust when controlling for other regulatory policies, bank and country characteristics. Regarding activity restrictions, we find that tighter activity restrictions are negatively associated with banking efficiency. By promoting the exchange of information, banks that engage in broader activities such as securities, insurance and real estate investment may benefit from economies of scale and scope, hence enhancing the efficiency levels. Similarly, we find that policies that promote official supervision may have adverse effects on banking efficiency. The results indicate that the official supervisory approach is ineffective in reducing overhead costs. However, the estimated coefficient is not significant enough to explain the behaviour of the net interest margin variable. The empirical evidence supports the private monitoring approach. Countries that require banks to disclose accurate and detailed information to the public tend to have a more efficient banking system. Basel II Capital Accord emphasises the importance of capital regulation, official supervisory power and private monitoring. Nevertheless, our findings incline to the negative aspects of capital regulation and official supervisory power.

Moreover, we find that bank-specific and country-specific control variables help explain banking efficiency. Larger banks with lower market share tend to have higher efficiency levels as confirmed by the significant negative coefficient of bank size and the significant positive coefficient of bank market share. Bank equity appears to be an important driver of banking efficiency only in the second model when overhead costs are used as a proxy. Similarly, government-owned banks variable only influences banking efficiency when considering overhead costs. This finding supports that countries that promote government-owned banks are more effective in reducing overhead costs, thus improving banking efficiency. Additionally, the growth rate of GDP per capita is negatively associated with net interest margin and overhead costs. In this perspective, the banking sector in expanding countries is significantly more efficient.

Our evidence overall suggests that the link between capital regulation and banking efficiency as measured by net interest margin and overhead costs is not robust enough to control for other regulatory variables. The results confirm that activity restrictions have a negative and significant impact on banking efficiency. Also, policies that encourage official supervisory power do not enhance the efficiency of the banking sector. The only approach that is positively and statistically significantly associated with efficiency is private monitoring.

This study may be subject to some limitations, therefore our findings should be interpreted carefully. First, we used cross-section data instead of panel data since information on regulatory measures is collected from the last survey conducted by the World Bank. Thus, the data is only available for one point in time (i.e. Bank, 2010). It is argued that banking efficiency is better estimated with panel data (Coelli et al., 2005). Another limitation of this study is the measures of banking efficiency. The literature on banking efficiency highlights the advantages of efficient frontier approaches compared to traditional accounting ratios (Berger & Humphrey, 1997; Fethi & Pasiouras, 2010). Besides, we consider 21 EU countries. Due to data unavailability, we do not consider all EU countries. As explained in the third section, countries economically large as Germany and the United Kingdom are not included, which could influence our results.

6. Recommendations

Given the negative relationship found between banking efficiency and official supervisory power, the recommendation for policy makers is to re-direct resources away from official supervision and towards private monitoring, a concept that has been promulgated in the Basel II framework. Moreover, supervision should not focus on restricting activities, as this is seen to have a significant and negative impact on banking efficiency. Supervisors should also recognize that, while capital regulation can be important for individual banks, it does very little for banking efficiency, but government-owned banks can contribute to system efficiency.

There are other regulatory policies than those considered in this study, such as financial conglomerate variables, competition, regulatory variables and deposit insurance scheme, which could affect banking efficiency. Larger sample size and additional regulatory variables could be considered. To further explore implications of regulatory policies, several characteristics of the banking system can be studied such as loan growth and bank competition.

Given the vast scope of banking regulation, many aspects can be explored in the future.

Funding

The authors received no direct funding for this research.

Author details

Edward Bace¹ E-mail: E.Bace@mdx.ac.uk ORCID ID: http://orcid.org/0000-0003-4452-0350 Ana Ferreira¹ E-mail: am3150@live.mdx.ac.uk ¹ Accounting & Finance, Middlesex University, London, UK.

Citation information

Cite this article as: Regulation's influence on EU banking efficiency: An evaluation post crisis, Edward Bace & Ana Ferreira, *Cogent Economics & Finance* (2020), 8: 1838735.

Notes

- The study uses inefficiency as proxy for banking efficiency in the DEA approach. Instead of measuring how close the bank is from the non-parametric efficient frontier, the study measures how far it is. By definition, the best practice banks are 100% efficient, therefore inefficiency equals the difference between 100 and the efficiency score.
- 2. The definition of financial conglomerates adopted in this paper is the same proposed by Laeven and Levine (2007).
- 3. See, for example, King and Levine (1993) and Levine (1997) for empirical evidence.
- 4. The study separated financial crisis into three categories: banking crisis, currency crisis and debt crisis. ⁵ Bank charter value is defined as the value that would be foregone in case of closure. The study measures bank charter value with Tobin's Q ratio.
- The 28 EU countries are: Austria, Belgium, Bulgaria, Croatia, Cyprus, Czech Republic, Denmark, Estonia, Finland, France, Germany, Greece, Hungary, Ireland, Italy, Latvia, Lithuania, Luxembourg, Malta, Netherlands, Poland, Portugal, Romania, Slovakia, Slovenia, Spain, Sweden and United Kingdom.

References

- Altunbas, Y., Carbo, S., Gardener, E. P., & Molyneux, P. (2007). Examining the relationships between capital, risk and efficiency in European banking. *European Financial Management*, 13(1), 49–70. https://doi.org/ 10.1111/j.1468-036X.2006.00285.x
- Aron, D. J. (1988). Ability, moral hazard, firm size, and diversification. The Rand Journal of Economics, 19(1), 72–87. https://doi.org/10.2307/2555398
- Barth, J. R., Caprio, G., Jr., & Levine, R. (2004). Bank regulation and supervision: What works best. Journal of Financial Intermediation, 13(2), 205–248. https://doi. org/10.1016/j.jfi.2003.06.002
- Barth, J. R., Caprio, G., Jr., & Levine, R. (2008). Bank regulations are changing: For better or worse? *Comparative Economic Studies*, 50(4), 537–563. https://doi.org/10.1057/ces.2008.33
- Barth, J. R., Caprio, G., Jr., & Levine, R. (2013a). Bank regulation and supervision in 180 countries from 1999 to 2011. *Journal of Financial Economic Policy*, 5(2), 111–219. https://doi.org/10.1108/17576381311329661
- Barth, J. R., Caprio Jr., G., & Levine, R. (2006). Rethinking bank regulation: Till angels govern. Cambridge University Press.

- Barth, J. R., Lin, C., Ma, Y., Seade, J., & Song, F. M. (2013b). Do bank regulation, supervision and monitoring enhance or impede bank efficiency? *Journal of Banking & Finance*, 37(8), 2879–2892. https://doi.org/ 10.1016/j.jbankfin.2013.04.030
- Beck, T., Demirguc-Kunt, A., & Levine, R. (2006a). Bank concentration, competition, and crises: First results. *Journal of Banking & Finance*, 30(5), 1581–1603. https://doi.org/10.1016/j.jbankfin.2005.05.010
- Beck, T., Demirguc-Kunt, A., & Levine, R. (2006b). Bank supervision and corruption in lending. *Journal of Monetary Economics*, 53(8), 2131–2163. https://doi. org/10.1016/j.jmoneco.2005.10.014
- Berger, A. N. (1995). The relationship between capital and earnings in banking. *Journal of Money, Credit and Banking*, 27(2), 432–456. https://doi.org/10.2307/ 2077877
- Berger, A. N., & Humphrey, D. B. (1997). Efficiency of financial institutions: International survey and directions for future research. European Journal of Operational Research, 98(2), 175–212. https://doi.org/10.1016/ S0377-2217(96)00342-6
- Besanko, D., & Kanatas, G. (1996). The regulation of bank capital: Do capital standards promote bank safety? *Journal of Financial Intermediation*, 5(2), 160–183. https://doi.org/10.1006/jfin.1996.0009
- Blum, J. (1999). Do capital adequacy requirements reduce risk in banking? Journal of Banking & Finance, 23(5), 755–771. https://doi.org/10.1016/S0378-4266(98) 00113-7
- Boot, A. W., & Thakor, A. V. (1993). Self-interested bank regulation. American Economic Review, 83(2), 206–212. https://www.jstor.org/stable/2117665?seq=1
- Boyd, J. H., Chang, C., & Smith, B. D. (1998). Moral hazard under commercial and universal banking. Journal of Money, Credit and Banking, 30(3), 426–468. https:// doi.org/10.2307/2601249
- Chortareas, G. E., Girardone, C., & Ventouri, A. (2012). Bank supervision, regulation, and efficiency: Evidence from the European Union. *Journal of Financial Stability*, 8(4), 292–302. https://doi.org/10.1016/j.jfs. 2011.12.001
- Cihak, M., Demirguc-Kunt, A., Peria, M. S., & Mohseni-Cheraghlou, A. (2012). Bank regulation and supervision around the world: A crisis update (Working Paper No 6286). World Bank Policy Research.
- Coelli, T. J., Rao, D. S., O'Donnell, C. J., & Battese, G. E. (2005). An introduction to efficiency and productivity analysis (2nd ed.). Springer US.
- Deli, Y. D., & Hasan, I. (2017). Real effects of bank capital regulations: Global evidence. Journal of Banking & Finance, 82(9), 217–228. https://doi.org/10.1016/j. jbankfin.2016.11.022
- Demirguc-Kunt, A., Enrica, D., & Tressel, T. (2008). Banking on the principles: Compliance with basel core principles and bank soundness. *Journal of Financial Intermediation*, 17(4), 511–542. https://doi.org/10. 1016/j.jfi.2007.10.003
- Demirguc-Kunt, A., Laeven, L., & Levine, R. (2004). Regulations, market structure, institutions, and the cost of financial intermediation. *Journal of Money, Credit & Banking*, 36(3), 592–622. https://doi.org/10. 1353/mcb.2004.0045
- Diamond, D. W. (1991). Monitoring and reputation: The choice between bank loans and directly placed debt.

Journal of Political Economy, 99(4), 689–721. https:// doi.org/10.1086/261775

- Djankov, S., La Porta, R., Lopez-de-Silanes, F., & Shleifer, A. (2002). The regulation of entry. *The Quarterly Journal of Economics*, 117(1), 1–37. https://doi.org/10.1162/ 003355302753399436
- Duarte, J., Han, X., Harford, J., & Young, L. (2008). Information asymmetry, information dissemination and the effect of regulation FD on the cost of capital. *Journal of Financial Economics*, 87(1), 24–44. https:// doi.org/10.1016/j.jfineco.2006.12.005
- Fernandez, A. I., & Gonzalez, F. (2005). How accounting and auditing systems can counteract risk-shifting of safety-nets in banking: Some international evidence. *Journal of Financial Stability*, 1(4), 466–500. http:// www.sciencedirect.com/science/article/pii/S1572-3089(05)00026-4
- Fethi, M. D., & Pasiouras, F. (2010). Assessing bank efficiency and performance with operational research and artificial intelligence techniques: A survey. European Journal of Operational Research, 204(2), 189–198. https://doi.org/10.1016/j.ejor. 2009.08.003
- Furlong, F. T., & Keeley, M. C. (1989). Capital regulation and bank risk-taking: A note. Journal of Banking & Finance, 13(6), 883–891. https://doi.org/10.1016/ 0378-4266(89)90008-3
- Gonzalez, F. (2005). Bank regulation and risk-taking incentives: An international comparison of bank risk. *Journal of Banking & Finance*, 29(5), 1153–1184. https://doi.org/10.1016/j.jbankfin.2004.05.029
- Hay, J. R., & Shleifer, A. (1998). Private enforcement of public laws: A theory of legal reform. The American Economic Review, 88(2), 398–403. https://scholar.har vard.edu/shleifer/publications/private-enforcementpublic-laws-theory-legal-reform
- Jensen, M. C. (1986). Agency costs of free cash flow, corporate finance, and takeovers. The American Economic Review, 76(2), 323–329. https://www.jstor. org/stable/1818789?seq=1
- Kaufman, G. G. (1992). Capital in banking: Past, present and future. Journal of Financial Services Research, 5 (4), 385–402. https://doi.org/10.1007/BF00114074
- Keeley, M. C., & Furlong, F. T. (1990). A reexamination of mean-variance analysis of bank capital regulation. *Journal of Banking & Finance*, 14(1), 69–84. http:// www.sciencedirect.com/science/article/pii/0378-4266(90)90036-2
- Kim, T., Koo, B., & Park, M. (2013). Role of financial regulation and innovation in the financial crisis. *Journal* of Financial Stability, 9(4), 662–672. https://doi.org/ 10.1016/j.jfs.2012.07.002
- King, R. G., & Levine, R. (1993). Finance and growth: Schumpeter might be right. The Quarterly Journal of Economics, 108(3), 717–737. https://doi.org/10.2307/ 2118406
- Koehn, M., & Santomero, A. M. (1980). Regulation of bank capital and portfolio risk. The Journal of Finance, 35 (5), 1235–1244. https://doi.org/10.1111/j.1540-6261. 1980.tb02206.x
- Kopecky, K. J., & VanHoose, D. (2006). Capital regulation, heterogeneous monitoring costs, and aggregate loan quality. *Journal of Banking & Finance*, 30(8), 2235–2255. https://doi.org/10.1016/j.jbankfin.2005.07.018
- La Porta, R., Lopez-de-Silanes, F., & Shleifer, A. (2002). Government ownership of banks. *The Journal of*

Finance, 57(1), 265–301. https://doi.org/10.1111/ 1540-6261.00422

- Laeven, L., & Levine, R. (2007). Is there a diversification discount in financial conglomerates? *Journal of Financial Economics*, 85(2), 331–367. https://doi.org/ 10.1016/j.jfineco.2005.06.001
- Laeven, L., & Levine, R. (2009). Bank governance, regulation and risk taking. *Journal of Financial Economics*, 93(2), 259–275. https://doi.org/10.1016/j.jfineco. 2008.09.003
- Levine, R. (1997). Financial development and economic growth: Views and Agenda. *Journal of Economic Literature*, 35(2), 688–726. https://www.jstor.org/ stable/2729790?seq=1
- Levine, R. (2005). The microeconomic effects of different approaches to bank supervision. In Conference on economics, political institutions, and financial markets II: Institutional theory and evidence from Europe, the United States, and Latin America. Stanford, California: Stanford University.
- Pasiouras, F., Gaganis, C., & Zopounidis, C. (2006). The impact of bank regulations, supervision, market structure, and bank characteristics on individual bank ratings: A cross-country analysis. *Review of Quantitative Finance and Accounting*, 27(4), 403438. https://doi.org/10.1007/s11156-006-0045-0
- Pasiouras, F., Tanna, S., & Zopounidis, C. (2009). The impact of banking regulations on banks' cost and profit efficiency: Cross-country evidence. International Review of Financial Analysis, 18(5), 294–302. https://doi.org/10.1016/j.irfa.2009.07.003
- Quintyn, M., & Taylor, M. W. (2003). Regulatory and supervisory independence and financial stability. CESifo Economic Studies, 49(2), 259–294. https://doi. org/10.1093/cesifo/49.2.259
- Rajan, R. G. (1992). Insiders and outsiders: The choice between informed and arm's-length debt. The Journal of Finance, 47(4), 1367–1400. https://doi.org/ 10.1111/j.1540-6261.1992.tb04662.x
- Rotemberg, J. J., & Saloner, G. (1994). Benefits of narrow business strategies. The American Economic Review, 84(5), 1330–1349. https://www.jstor.org/stable/ 2117775?seq=1
- Santos, J. A. (2001). Bank capital regulation in contemporary banking theory: A review of the literature. *Financial Markets, Institutions & Instruments, 10*(2), 41–84. https://doi.org/10.1111/1468-0416.00042
- Shleifer, A., & Vishny, R. W. (1998). The grabbing hand: Government pathologies and their cures (18th ed.). Harvard University Press.
- Stein, J. C. (2002). Information production and capital allocation: Decentralized versus hierarchical firms. *The Journal of Finance*, 57(5), 1891–1921. https://doi. org/10.1111/0022-1082.00483
- VanHoose, D. (2007). Theories of bank behavior under capital regulation. *Journal of Banking & Finance*, 31 (12), 3680–3697. https://doi.org/10.1016/j.jbankfin. 2007.01.015
- World Bank (2010). World development indicators. https://databank.worldbank.org/source/world-development-indicators
- Yildirim, S. H., & Philippatos, G. C. (2007). Efficiency of banks: Recent evidence from the transition economies of Europe, 1993–2000. European Journal of Finance, 13(2), 123–143. https://doi.org/10.1080/ 13518470600763687



© 2020 The Author(s). This open access article is distributed under a Creative Commons Attribution (CC-BY) 4.0 license.

You are free to:

Share — copy and redistribute the material in any medium or format. Adapt — remix, transform, and build upon the material for any purpose, even commercially. The licensor cannot revoke these freedoms as long as you follow the license terms. Under the following terms: Attribution — You must give appropriate credit, provide a link to the license, and indicate if changes were made. You may do so in any reasonable manner, but not in any way that suggests the licensor endorses you or your use. No additional restrictions You may not apply legal terms or technological measures that legally restrict others from doing anything the license permits.

Cogent Economics & Finance (ISSN: 2332-2039) is published by Cogent OA, part of Taylor & Francis Group. Publishing with Cogent OA ensures:

- Immediate, universal access to your article on publication
- High visibility and discoverability via the Cogent OA website as well as Taylor & Francis Online
- Download and citation statistics for your article
- Rapid online publication
- Input from, and dialog with, expert editors and editorial boards
- Retention of full copyright of your article
- Guaranteed legacy preservation of your article
- Discounts and waivers for authors in developing regions

Submit your manuscript to a Cogent OA journal at www.CogentOA.com